Author’s response to reviews

Title: PRegnancy Outcomes after a Maternity Intervention for Stressful EmotionS (PROMISES): study protocol for a randomised controlled trial

Authors:

Judith L Meijer (j.l.meijer@epi.umcg.nl)
Claudi LH Bockting (c.l.h.bockting@rug.nl)
Chantal Beijers (c.beijers@epi.umcg.nl)
Tjitte Verbeek (t.verbeek@epi.umcg.nl)
A. Dennis Stant (a.d.stant@epi.umcg.nl)
Johan Ormel (j.ormel@med.umcg.nl)
Ronald P Stolk (r.p.stolk@epi.umcg.nl)
Peter de Jonge (p.de.jonge@med.umcg.nl)
Mariëlle G van Pampus (m.g.vanpampus@olvg.nl)
Huibert Burger (h.burger@epi.umcg.nl)

Version: 5 Date: 8 June 2011

Author’s response to reviews: see over
Dear Editors,

Hereby we send you a revised version of our manuscript entitled ‘PRegnancy Outcomes after a Maternity Intervention for Stressful EmotionS (PROMISES): study protocol for a randomised controlled trial’ for publication as an original article in Trials.

- We made the required changes on reference 32.
- We checked the manuscript for typographical errors.

We hope that you will find the revised manuscript suitable for publication in Trials.